Author’s response to reviews

Title: Availability of Adequately Iodized Salt in Northwest Ethiopia: a Cross-sectional Study

Authors:
Zegeye Abebe (zegeye24@gmail.com)
Amare Tariku (amaretariku15@yahoo.com)
Ejigu Gebeye (ejigugebeye@gmail.com)

Version: 2 Date: 05 May 2017

Author’s response to reviews:

Author's response to reviews

Title: Availability of Adequately Iodized Salt in Northwest Ethiopia: a Cross-sectional Study

Authors:
Zegeye Abebe (ZA): zegeye24@gmail.com
Amare Tariku (AT): amaretariku15@yahoo.com
Ejigu Gebeye (EG): ejigugebeye@gmail.com

Version: 2 Date: May, 2017

With regards!

From: Zegeye Abebe (E-mail: zegeye24@gmail.com)

Correspondence Author
To: Archives of Public Health Journal, Editorial Office

Subject: Submitting a revised version of manuscript

Object: Manuscript ID= AOPH-D-16-00152: Availability of Adequately Iodized Salt in Northwest Ethiopia: a Cross-sectional Study

We would like to thank the reviewer and editor for their unreserved support. We also found the comments are very important which will improve clarity of our manuscript. After we go through the whole document, the point-by-point responses for each of the comments and the revised manuscript with track changes are provided in the attached documents.

Point by point Response

Reviewer #2

Comments and Responses

1. P3L18. "Iodine deficiency is the major public health problem, especially for children and pregnant women (3)". Please correct "is a major public health problem".

Answer: Dear reviewer, thank you very much for your invaluable comments!!! The comment is accepted

2. P4L26 "Even though, the country Ethiopia is salt producing state, planned to reach utilization of iodized salt >90% by the year 2015 (21). Please correct "is in a salt…”

Answer: Thank you the comment is accepted

3. P9L10 The current report is higher .." what is higher? The current report shows that iodized salt use is higher in… (indicate the percentage) than in…

Answer: The comment is accepted and corrected in the main document

Thank you very much!!!